Author's response to reviews

Title: The Use of Complementary and Alternative Medicine Products in Preceding Two Days Among Finnish Parents - a population survey

Authors:

Katri P Hämeen-Anttila (katri.hameen-anttila@fimea.fi)
Ulla R Niskala (ulla.niskala@apteekit.net)
Sanna M Siponen (sanna.siponen@uef.fi)
Riitta S Ahonen (riitta.ahonen@uef.fi)

Version: 4 Date: 3 November 2011

Author's response to reviews: see over
BMC Complementary and Alternative Medicine

Dear Editor,

Please find enclosed our final manuscript "The Use of Complementary and Alternative Medicine Products in Preceding Two Days Among Finnish Parents – a population survey” which has been accepted for publication in BMC Complementary and Alternative Medicine as research article.

All authors are responsible for the reported research. They all have participated in the concept and design, analysis and interpretation of data, drafting or revising of the manuscript, and they have approved the manuscript as submitted. None of the authors has any professional affiliation, financial agreement, or other involvement with any company that might constitute a conflict of interest regarding this manuscript.

Changes made in the final proofs state are:
• The abstract has been changed to be identical in the manuscript and the submission system
• For reference number 22 has been provided a functioning link
• Reference number 28 has been updated for bibliographical information
• Highlighting and tracking has been removed from the manuscript
• Manuscript has been checked for typographical and other errors.

Yours sincerely,

Hämeen-Anttila K\textsuperscript{a}, PhD, development manager
Niskala U\textsuperscript{b} MSc (Pharm)
Siponen S\textsuperscript{b}, MSc (Pharm), researcher
Ahonen R\textsuperscript{b}, PhD, professor

\textsuperscript{a} Finnish Medicines Agency, Helsinki, Finland
\textsuperscript{b} School of Pharmacy, University of Eastern Finland, Kuopio, Finland

Address for correspondence:
Katri Hämeen-Anttila, Adjunct professor, Ph.D.(pharm.)
Development manager
Finnish Medicines Agency
Mikokatu 1, Kuopio, P.O. Box 55
FI-00301 Helsinki, FINLAND
Gsm. +358 505773075
E-mail. katri.hameen-anttila[at]fimea.fi

Kuopio 2\textsuperscript{nd} November 2011